

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Schneider, Martin

Working Paper What contributes to consumer price inflation? A novel decomposition framework with an application to Austria

Working Paper, No. 255

Provided in Cooperation with: Oesterreichische Nationalbank (OeNB), Vienna

Suggested Citation: Schneider, Martin (2024) : What contributes to consumer price inflation? A novel decomposition framework with an application to Austria, Working Paper, No. 255, Oesterreichische Nationalbank (OeNB), Vienna

This Version is available at: https://hdl.handle.net/10419/289805

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



WORKING PAPER 255

What contributes to consumer price inflation? A novel decomposition framework with an application to Austria

Martin Schneider

The Working Paper series of the Oesterreichische Nationalbank is designed to disseminate and to provide a platform for discussion of either work of the staff of the OeNB economists or outside contributors on topics which are of special interest to the OeNB. To ensure the high quality of their content, the contributions are subjected to an international refereeing process. The opinions are strictly those of the authors and do in no way commit the OeNB.

The Working Papers are also available on our website (http://www.oenb.at) and they are indexed in RePEc (http://repec.org/).

Publisher and editor	Oesterreichische Nationalbank Otto-Wagner-Platz 3, 1090 Vienna, Austria PO Box 61, 1011 Vienna, Austria www.oenb.at oenb.info@oenb.at Phone (+43-1) 40420-046698
Editor	Martin Summer
Cover Design	Information Management and Services Division
DVR 0031577	
ISSN 2310-5321 (Print)	

ISSN 2310-533X (Online)

© Oesterreichische Nationalbank, 2024. All rights reserved.

What contributes to consumer price inflation? A novel decomposition framework with an application to Austria¹

Martin Schneider (Oesterreichische Nationalbank)

Abstract:

In this paper, we propose a new methodology for decomposing consumer price inflation into contributions of cost components using national accounts data. It builds on the well-known decomposition method for the value-added deflator and expands it by combining the cost structure of the consumption bundle underlying the harmonized index of consumer prices (HICP) derived from an input-output table with quarterly national accounts data. This allows to decompose HICP inflation into detailed cost components including imports. We apply the approach to Austria and analyze the composition of inflation for the period from the first quarter of 2019 to the first quarter of 2023. In 2022, the most significant contributors to inflation were both energy and non-energy imports. Profits contributed to inflation from the second half of 2022 onwards, whilst there have been no substantial price pressures from wages. We also find that there are considerable differences in the inflation determinants between subindices of the HICP. Whilst imports played a crucial role for inflation of food, non-energy industrial goods and energy, their influence for services inflation was minimal. The results of the analysis show that the decomposition can provide valuable insights for the conduct of monetary policy.

Keywords: Consumer price inflation, production-side decomposition, inflation accounting *JEL-Classification:* C67, E25, E31.

¹ <u>martin.schneider@oenb.at.</u> The views expressed in this paper are strictly those of the authors and do not necessarily reflect the views of the Oesterreichische Nationalbank. The author likes to thank Friedrich Fritzer, Erwin Kolleritsch, Zuzana Molnarova, Lukas Reiss, Mirjam Salish, Richard Sellner and Patricia Walter and the participants of an internal discussion round for helpful comments and suggestions.

Non-technical summary

The current phase of high inflation has led to lively interest in this topic. One question that was often asked concerns the relative role of wages and profits for inflation. Most papers that analyze this topic use a simple decomposition of the value-added deflator. In this paper we propose a novel methodology for a production-side decomposition of HICP inflation based on national accounts data. We build on the decomposition of the valueadded deflator and expand it by combining input-output data with sectoral quarterly national accounts data. This allows us to decompose HICP inflation at a granular level.

We use the approach to analyze the inflation developments in Austria for the period from the first quarter of 2019 to the first quarter of 2023. In 2022, energy as well as nonenergy imports were the most important contributors to inflation. Profits contributed to inflation from the second half of 2022 onwards, whilst there have been no substantial contributions from wages. We find that there are considerable differences in the inflation determinants between subindices of the HICP. Whilst imports played the main role for inflation of food, non-energy industrial goods and energy, they hardly contributed to services inflation. The results also clearly demonstrate how particularities like the lagged wage-bargaining process in Austria impact on inflation.

We also compare the decomposition of the HICP with the decomposition of the valueadded deflator, highlighting five key differences. Notable distinctions include the role of imports, the exclusion of taxes and subsidies in the value-added deflator, variations in sectoral compositions, different time series used, and the use of distinct quantities for unit cost components. These factors result in significant disparities in the relative contributions of wages and profits, with wages playing a stronger role for the value-added deflator. Additionally, we find opposite signs of profit contributions in 2021 due to divergent growth patterns between real consumption and real value added.

The paper also discusses the limitations of such a decomposition approach. It is a decomposition that is based on an accounting identity, and hence cannot explain the magnitude of inflation but only the distribution of incomes arising from the production of consumption goods. Causal relationships inferred from such identities can be misleading, as we demonstrate with the GDP demand side identity. We also discuss the potential of the decomposition in identifying the nature of inflationary shocks, distinguishing between demand-pull, cost-push, and monetary shocks. Additionally, we discuss the complexities in analyzing the role of profits in inflation, considering factors like market structure and measurement errors.

1. Introduction

Understanding inflation developments is of crucial importance for central banks. Given the importance of this topic, various approaches have been used to analyze the inflation process. One simple approach, which has been used extensively since at least two decades is a decomposition of the value-added deflator using national accounts data (ECB, 2003, ECB, 2006, Jaumotte and Morsy, 2012). The change of the value-added deflator is decomposed into contributions of the value-added components compensation of employees, net taxes on production, consumption of fixed capital and net operating surplus. This approach has become increasingly prominent during the current period of high inflation, see Abberger and Nierhaus (2023), Arce et al. (2023), Byrne et al. (2022), European Commission (2023), Fritzer et al. (2023), Hahn (2019), Hansen et al. (2023), Haskel (2023), Hebbing and Öztürk (2023) and Richardson et al. (2022) just to name a few of the numerous recent contributions in that field.²

Although this approach generates interesting insights into the structure of domestic valueadded inflation, it is not the optimal approach for central banks, which have their focus on consumer price inflation. For the conduct of monetary policy, it is crucial that a central bank understands the composition of consumer price inflation in detail, especially with regard to the extent to which domestic cost components vs. imports contribute to inflation.

Therefore, we propose a novel approach that enhances this simple approach in a way that allows for a decomposition of the Harmonized Index of Consumer Prices (HICP). It utilizes quarterly national accounts data in combination with the cost structure of consumption goods derived from the input-output table. This allows to decompose consumer price inflation at a granular level into contributions from unit cost components³ including domestic value-added components and imports.

Whilst there are some papers available that decompose consumer prices, this is the first paper that decompose HICP inflation in such a consistent and detailed way. Recently, there have been some papers that decompose growth of the consumption deflator into domestic contributions and imports. The European Commission (2023) analyses inflation and competitiveness developments for seven CESEE EU member countries. Besides other approaches, this paper utilizes a decomposition of the consumption deflator. The decomposition is very crude, since it decomposes the growth of the consumption deflator

² There numerous other approaches used to analyse the drivers of inflation. One popular approach utilises the Phillips curve. Under this approach, economic slack (e.g. the output gap) is the main driver of inflation, where an increase in slack is associated with a reduction in inflation (see e.g. Rumler, 2007, Stock and Watson, 2019, and Moretti et al., 2019). There are countless studies that analyse inflation developments with (structural) VAR models (see e.g. Aucremanne and Wouters, 1999, Wehinger, 2000, Martel, 2008, Kambar and Wong, 2018). Semi-structural models have also been heavily utilized to analyze inflation developments (see e.g. Guo et al. 2019). Another widely used approach of modelling the determinants of inflation is based on dynamic stochastic general equilibrium (DSGE) models in which inflation is mainly driven by the production costs of firms (see e.g. Smets and Wouters, 2003).

³ Unit cost components are defined as the nominal cost component (e.g. compensation of employees) divided by the quantity of the private consumption good.

into contributions stemming from imports (food, energy and other imports), a passthrough markup and a residual. The residual captures all domestic components. Hansen et al. (2023) decompose the consumption deflator for the euro area. Therefore, they build on the decomposition of sectoral value-added deflators into contributions of labor, profits and net taxes. The contribution of imports is defined as the residual between the change of the consumption deflator and the weighted sum of changes of sectoral value-added deflators.

For the time being, there are at least two papers that decompose the CPI into contributions stemming from domestic and imported sources. Haskel (2023) decomposes the CPI for the UK, US and the euro area. His decomposition is based on a decomposition of the private consumption deflator with National Accounts data plus a residual term that captures the differences between the CPI and the private consumption deflator. Dhingra (2023) decomposes the CPI for the UK with a decomposition that does not use data from annual or quarterly national accounts. Instead, she combines data from supply-use tables with time series data for wages and various prices.

One of the papers that use input-output tables to decompose other inflation measures is that of Eder et al. (2019). They calculate producer price differentials for nine European countries vis-à-vis Germany for four broad sectors and decompose them using the Leontief prices model and a structural decomposition approach (SDA).

The remainder of the paper is organized as follows. Section 2 presents the simple valueadded deflator decomposition as standard in the literature. Section 3 gives an overview of the methodology. In section 4, the calculation of cost components from the input-output table is presented. In section 5 we calculate inflation contributions by combining cost components with quarterly national accounts data. Section 6 discusses data issues and computational details. In section 7 we discuss important points to consider then analyzing the results. Section 8 presents the results of the decomposition for Austria. In section 9, we look at the differences between decomposition of the value-added deflator and the HICP. Finally, section 10 concludes.

2. A simple decomposition of the value-added deflator

We start with an illustrative example of a simple supply-side decomposition of the valueadded deflator of industry j ($p_{j,t}$) from national accounts (see ECB, 2006 and Jaumotte and Morsy, 2012 for early applications). Nominal value-added of industry j ($y_{j,t}^r p_{j,t}$) is defined as the sum of k nominal cost (income) components $CC_{k,j,t}$ (compensation of employees, other indirect taxes less subsidies on production, consumption of fixed capital and net operating surplus⁴ (including mixed income)).

$$y_{j,t}^{r} p_{j,t} = \sum_{k=1}^{K} c c_{k,j,t}$$
(1)

⁴ Profits (net operating surplus) are not a cost component in a literal sense from a firm's perspective. For the sake of simplicity, we refer to them as a cost component.

Dividing this equation by real value-added $y_{j,t}^r$, we obtain an equation which defines the value-added deflator as the sum of its per unit cost components ($ucc_{k,j,t}$).

$$p_{j,t} = \sum_{k=1}^{K} \frac{cc_{k,j,t}}{y_{j,t}^r} = \sum_{k=1}^{K} ucc_{k,j,t}$$
(2)

We now take the first difference and divide both side of the equation by the previous period's value-added deflator. In addition, we expand each term of the right-hand side by its previous period's unit costs. The percentage change of the sectoral output price is hence defined as the sum of the percentage changes of its unit cost components weighted with previous period's real unit cost component.

$$\frac{dp_{j,t}}{p_{j,t-1}} = \sum_{k=1}^{K} \frac{ducc_{k,j,t}}{ucc_{k,j,t-1}} \frac{ucc_{k,j,t-1}}{p_{j,t-1}}$$
(3)

Total economy Annual change in % (value added deflator) resp. contributions in percentage points 14 12 10 8 6 4 0 -2 -4 -6 '20Q2 '20Q4 '21Q2 '2104 2202 '2204 2302 Compensation of employees Taxes less subsidies Consumption of fixed capital Net operating surplus Deflator Source: Statistics Austria, author's calculations,

Figure 1: Decomposition of changes of the value-added deflator for Austria

Figure 1 shows the results of the simple decomposition for Austria. In 2022, the acceleration of the value-added deflator growth was mainly explained by net operating surplus. Compensation of employees also contributed to the increase of the deflator, but not to its acceleration.

3. Overview of the methodology

In this section, we outline our novel methodology that expands the simple decomposition approach introduced in section 2 in a way that allows to decompose HICP inflation. Before presenting an overview of the method, it is important to understand the conceptual differences between the value-added deflator and the HICP.

3.1 Conceptual differences between the value-added deflator and the HICP

The value-added deflator $p_{j,t}$ describes the price of domestic value-added generated by production of industry j (see the left part of figure 2). Value-added neither includes imports nor intermediate goods purchased from domestic firms. The output of industry j is sold to other firms as intermediate or investment goods, to private households, the government, and foreign customers. It is nominated at basic prices, i.e. excluding trade and transport margins and net taxes on products. Since domestic value-added of industry j does not include intermediate inputs purchased from other domestic firms or from abroad, changes in cost components of other industries do not spread over to industry j.⁵

Figure 2: Overview of the cost components of the value-added deflator vs. the HICP within the input-output table framework



Footnote: This figure illustrates the decompositions of the value-added deflator and of the consumer price index within the framework of a symmetric input-output table. Such a (simplified) input-output table consists of the matrix of intermediate demand (A), a matrix that contain final demand components as well as net taxes and direct imports and a matrix that contains value-added components and indirect imports. Columns refer to industries resp. net taxes, direct imports and final demand components. Rows refer to goods, value added components and imports. Green cells refer to the price index to be decomposed, orange-colored cells to the cost components which are the result the decomposition.

Source: Author's own draft.

The Harmonized Index of Consumer Prices (HICP) measures the price of a basket of goods and services acquired by households within a country for final consumption. This basket includes both domestically produced as well as directly imported goods and services. The right part of figure 2 shows the cost components of the production of consumption good *i*. In this case, we have to distinguish between *direct cost components* (direct imports and net taxes on production) and *indirect cost components*, which are generated by domestic production of consumption good *i*. Due to the inter-industry linkages generated by the use of intermediate goods, consumption good *i* also includes the value added by other

⁵ In contrast to the value-added deflator, the GDP deflator also includes taxes less subsidies on products.

(upstream) industries as well as imports. Hence, the indirect cost components include domestic value-added of all industries and indirect imports. The consumption goods that are purchased by private households are nominated at purchasing prices, i.e. they include trade and transport margins and net taxes on products.

The differences between the two price indices imply that the HICP decomposition requires substantial modifications to the decomposition method used for the value-added deflator. There are four main issues that must be addressed:

- There is no direct link between the aggregate consumption basket that is the basis of the HICP and national accounts aggregates. Hence, disaggregated HICP data must be used. We use data for 45 COICOP⁶ sub-groups.
- The COICOP-45 classification must be matched with national accounts data that are available at the CPA⁷-74 level for Austria.
- The breakdown of the income side of national accounts is available for *industries*, whereas consumer prices relate to *goods and services*. Hence, input-output tables must be used to capture the relationship between industries and goods.
- The use of different price concepts (basic vs. purchasing prices) implies that we must find a way to cope with trade and transport margins, which reflect the services provided by trade and transport sectors (see section 4.2).

3.2 Main parts of the analysis

The proposed approach utilizes quarterly national accounts data supplemented with additional data on imports and indirect taxes less subsidies in combination with the cost structure derived from input-output tables to decompose the HICP and its subindices. The method consists of two main parts. In the first part we calculate the cost shares for each consumption good from the input-output table at the CPA-74 level and aggregate them to the COICOP classification. In the second part, we update these cost components with data from quarterly national accounts (incl. own estimates for missing components) and trade data. Figure 3 gives a detailed overview of the calculations.

⁶ Classification of individual consumption by purpose.

⁷ Classification by activity.



Figure 3: Overview of the decomposition of HICP inflation

Source: Author's own draft.

4. Calculation of cost components from the input-output table

In this section we explain the calculation of cost components from the latest input-output table.

4.1 Calculation of cost components at the CPA-74 level

We use the input-output table for Austria for the year 2019 with 74 goods/industries.⁸ For each good *i*, we distinguish between *direct* and *indirect cost components*. *Direct cost components* are direct imports and taxes less subsidies on goods. These can be obtained from the input-output table for private consumption without further calculations. The inflation contribution from taxes on goods is a special case, as it can be calculated easily as the difference between HICP inflation and HICP inflation at constant tax rates as

⁸ Statistics Austria publishes an input-output table with 64 industries/goods on its webpage. A more detailed table with 74 industries/goods and several additional tables can be purchased (Statistik Austria, 2023). See section 6 for an overview and table A-3 for details on the tables we have used for the analysis.

published by Statistics Austria. Although we do not have to include direct taxes on goods in our cost components for the decomposition, we include it for reporting purposes.

Indirect cost components are generated by the domestic production of consumption good *i*. We use a multiplier analysis to derive the indirect cost component *k* in industry *j* attributable to the production of consumption good *i* ($cc_{k,j,i}^{CPA}$) at the CPA-74 level. For each good *i*, we set the final demand vector D_i to zero except for good *i*, which we set to the value of private consumption at basic prices from the input-output table. Multiplying D_i with the Leontief-inverse $((I - A)^{-1})^9$ gives the vector of output Q_i generated in all industries by the production of D_i .

$$Q_i = (I - A)^{-1} D_i (4)$$

The cost component k of industry j necessary to produce consumption good i $(cc_{k,j,i}^{CPA})$ can be derived by multiplying output of industry j generated by the production of consumer good i $(q_{j,i})$ with the share of cost component k $(cc_{k,j}^{IO})$ in output of industry j $(q_{j,i})$ from the input-output table.

$$cc_{k,j,i}^{CPA} = \frac{cc_{k,j}^{IO}}{q_i} q_{j,i}$$
(5)

For each of the 74 CPA goods i at basic prices, we obtain detailed indirect cost components for all 74 industries. These include compensation of employees, other indirect taxes less subsidies on production, consumption of fixed capital and net operating surplus (including mixed income), net taxes on intermediate demand as well as imports for 74 goods and services categories. This gives us a total of 5846 (=74*79) cost components for each CPA good. To keep the decomposition manageable, we aggregate these cost components in a meaningful way. With regards to the number of industries, we aggregate them to the 13 industries that are available in quarterly national accounts.¹⁰ The 74 import goods per industry *j* that are available in the input-output table are first aggregated to 25 goods and services categories. In addition, we sum up imports over all industries *j*, giving us 25 import cost components. We further sum up the components of net taxes on intermediate goods, and the two direct cost components taxes on goods and subsidies on goods, each to one overall component per good *i*. Direct imports are aggregated to the 25 goods and services categories as indirect imports and are added to indirect imports. This gives us a total of 80 cost components for each consumption good *i* at purchasing prices (=4*13 cost components per industry + 25 imports + 3 direct cost

⁹ A is the matrix of technology coefficients which can be obtained by dividing the matrix of intermediate demand by output by industry.

¹⁰ Agriculture, forestry and fishing (NACE A), Mining and quarrying, energy, water and sewerage (NACE BDE), Manufacturing (NACE C), Construction (NACE F), Trade and repair of motor vehicles (NACE G), Transportation and storage (NACE H), Accommodation and food services (NACE I), Information and communication (NACE J), Finance and insurance activities (NACE K), Real estate activities (NACE L), Business-related services (NACE MN), Public sector activities (NACE OPQ), Arts, entertainment, recreation and other private services (NACE RTU).

components). We collect these cost components for all CPA-74 goods in one matrix with the dimension 74x80 (see figure 4).¹¹

Figure 4: Cost components for each CPA-74 good derived from the inputoutput table



Source: Author's own draft.

4.2 Aggregation of cost components from CPA-74 to the COICOP-45 level

The next step is to aggregate the cost components from the CPA-74 level to the COICOP-45 level. Therefore, we need a correspondence table that contains private consumption for both classifications. Statistics Austria provides such a table (table 25) in its detailed input-output data set (Statistik Austria, 2023).¹² The correspondence table X^{PP} is nominated at purchasing prices, i.e. including trade and transport margins and net taxes on products. Table A-1 in Appendix 1 illustrates the structure of this table. Each row describes how consumption good *i* at the CPA-74 level can be mapped to one or more COICOP-45 categories *j*. To obtain an aggregation matrix \hat{X}^{PP} , we divide each element of the correspondence table X^{PP} by its row sum. See appendix 1 for some details on aggregation matrices.

$$\hat{x}_{i,j}^{PP} = \frac{x_{i,j}^{PP}}{\sum_{j=1}^{J} x_{i,j}^{PP}}$$
(6)

The aggregation of the direct cost components subsidies on products (sp^{CC45}) and direct imports of consumption goods (md^{CC45}) at the COICOP-45 level can be obtained by

$$sp^{CC45} = \left(\hat{X}^{PP'} * sp^{CPA74}\right)' \tag{7}$$

¹¹ In the remainder of the paper, we use subscript k for all cost components instead of cost components per industry j as we did in equation 5.

¹² For countries where the statistical office does not provide this table, it can be estimated with the RAS method utilizing consumption vectors for the COICOP and the CPA classifications and a seed matrix (see Cai and Vandyck, 2020).

$$md^{CC45} = \left(\hat{X}^{PP'} * md^{CPA74}\right)' \tag{8}$$

For the aggregation of indirect cost components, we need to address the issue caused by the use of different price concepts in the input output table. Goods purchased by consumers are at purchasing prices, i.e. they include trade and transport margins. This means the output of the trade and transport sectors is included in the price of the goods they purchase and that consumers do not consume wholesale or retail goods directly. In contrast, the indirect cost components that we have calculated are at basic prices, i.e. trade and transport sectors are reported as entities. If we look at correspondence table A-1, we see that the output of transport and trade sectors is zero, as it is included in the final value of other goods.¹³ This implies that the indirect cost components of these industries would be lost during aggregation when we use aggregation matrix X^{PP} . Hence, we must convert matrix X^{PP} from purchasing to basic prices (X^{BP}). Therefore, we take a look at the composition of domestically produced private consumption good *i* at purchasing prices (c_i^{DPP}) , which is defined as domestically produced private consumption good *i* at basic prices (c_i^{DBP}) plus wholesale trade margins (wtm_i) , retail trade margins (rtm_i) , transport margins (trm_i) , and taxes on products (tp_i) less subsidies on products $(sp_i).^{14}$

$$c_i^{DPP} = c_i^{DBP} + wtm_i + rtm_i + trm_i + tp_i - sp_i$$
(9)

We convert X^{PP} to basic prices in the following way. For each of the CPA-74 goods and for each margin (wholesale, retail, and transport), we deduct the share of the respective margin from the row in X¹⁵ and add the margin to the corresponding margin goods.¹⁶ Table A-2 shows the resulting correspondence table at basic prices (X^{BP}). We can see that now there are non-zero elements in the rows of the margin goods, i.e. the cost components of the trade and transport sectors are assigned to COICOP-45 good which include these margins in their final value. Multiplying the transpose aggregation matrix with the matrix of cost components at the CPA-74 level gives us the transpose matrix of cost components at the COICIP-45 level.¹⁷

$$cc_i^{IO45} = \left(X^{\rm BP'} * cc_i^{IO74}\right)' \tag{10}$$

We proceed by adding the direct cost components to the corresponding columns of CC_i^{I045} .

¹³ For transport margins there are some non-zero elements in the rows since households consume some transport services directly. For wholesale and retail trade, all elements in the respective row of X^{PP} are zero.

¹⁴ Margins and net taxes are available for each final demand component in the Austrian input-output table.

¹⁵ Therefore, we implicitly assume that the share of each margin in the value of each COICOP good is constant. ¹⁶ Wholesale trade margins are provided by wholesale- a. retail trade, repair of motor vehicles (45) and wholesale trade, excl. motor vehicles and -cycles (46), retail trade margins by retail trade, exc. motor vehicles and -cycles (47) and trade margins by Land transport services a. transport services via pipelines (49), water transport services (50), air transport services (51), warehousing and support services for transportation (52) and insurance, reinsurance and pension funding services (65). The shares of each margin goods in total margins per type of margin (wholesale, retail, and trade) can be obtained from the detailed set of input-output tables provided by Statistics Austria.

¹⁷ We apply this aggregation for indirect cost components only. Direct cost components are aggregated with the aggregation matrix at purchasing prices.

The next step is to convert the cost components into cost shares that sum up to one by dividing the cost components by private consumption at purchasing prices.

$$cs_i^{IO45} = \frac{cc_i^{IO45}}{c_i^{PP}}$$
 (11)

Finally, we aggregate the cost shares at the COICOP-45 level to headline inflation, four special aggregates (food, non-energy industrial goods, energy and inflation) and to twelve divisions by multiplying them with a matrix with the respective HICP weights (see appendix A-1).

$$cs_i^{IO17} = cs_i^{IO45} * \mathbf{W}^{HICP}$$
(12)

We add these 17 additional rows to the matrix of cost shares, resulting in a 62x80 matrix of cost shares.

$$cs_i^{I062} = cs_i^{I017} \mid cs_i^{I045} \tag{13}$$

The left panel of figure 5 shows the resulting cost shares of headline inflation and the special aggregates for the input-output table 2019. There are substantial differences between these aggregates. The import content is highest for non-energy industrial goods and lowest for services. For energy it is also rather low due to the high role of net taxes, which amount to more than half of the total value of energy goods. The share of wages is lowest for energy and highest for services. The role of profits is highest for services, partly due to the role of mixed income of self-employees, which is part of net operating surplus in national accounts. Net taxes also play an important role for consumer prices.

The right panel of figure 5 compares the cost structure of headline inflation for 2019 with the cost structure derived from the input-output table 2015. It shows that the differences are very small.¹⁸ In addition, it shows the cost structure of the value-added deflator directly calculated from national accounts data for these two years. The difference between the cost structure of the HICP and the value-added deflator will be discussed in section 9.

¹⁸ The similarity of the cost structures for those two years translates into very similar decompositions. For the year 2022, the inflation contribution of compensation of employees – which amounts to 1.2 percentage points based on the IO table 2019 – changes to 1.1 percentage points for the IO table 2015. Similarly, the changes for the other cost components are also very small (Net operations surplus: 2.0 pp \rightarrow 2.1 pp., Imports: 4.8 pp \rightarrow 4.6 pp).



Figure 5: Cost shares of HICP and the value-added deflator

5. Calculation of inflation contributions

In the second part of the exercise, we use quarterly national accounts data and trade data together with the cost structure of the 45 COICOP goods derived from the input-output table to decompose consumer price inflation. This part consists of four steps: the first step is to estimate the cost components that are missing in quarterly national accounts. In the second step, we isolate the share of the cost components implied by private consumption from other final uses and refer to them as consumption-implied cost components. The third step is to combine the cost components derived from the input-output table with growth rates of our consumption-implied cost components at the quarterly level. Finally, the last step is to decompose HICP inflation.

5.1 Estimation of missing quarterly national accounts components

We calculate inflation contributions based on quarterly national accounts (QNA) data per industry (plus detailed data on imports of goods and services). The published QNA data only contain value-added and compensation of employees per industry. Therefore, we must estimate the three missing components (net taxes on production, consumption of fixed capital and net operating surplus). The following sections briefly summarize the procedure. Details of the procedure can be found in Fritzer, Reiss and Schneider (2023).

Taxes less subsidies on production

In normal times, subsidies to firms do not play a significant role (apart from agriculture), as their share is low (2000-2019: 1.7% of gross value-added) and stable over time. In 2020 and 2021, however, their share increased to 7.1% due to the COVID pandemic. In 2022, the Austrian government paid subsidies to firms due to high energy prices. Fritzer, Reiss and Schneider (2023) use several data sources to break down quarterly subsidies by industry. The breakdown of short-time work subsidies is derived from detailed data from the Ministry of Labour. For the other crisis subsidies, the estimates are based on data

from the COVID-19 Federal Financing Agency (COFAG), the Ministry of Finance and Statistics Austria.

Consumption of fixed capital

Consumption of fixed capital or depreciation is currently available in annual national accounts up to 2021. To obtain quarterly estimates until 2023Q1, we first have to estimate annual values for 2022 and 2023 based on the perpetual inventory method. The capital real stock of industry j in year t ($k_{j,t}$) results from the capital stock of the previous year ($k_{j,t-1}$) plus real investments in year t ($i_{j,t}$) minus real depreciation ($d_{j,t}$). The capital stock of t-1 is depreciated with the depreciation rate r_j ; the investments of year t at half the rate (Huber, 2015).

$$k_{j,t} = k_{j,t-1} + i_t - d_t = k_{j,t-1} * (1 - r_j) + i_{j,t} * (1 - 0.5 * r_j)$$
(14)

Real depreciation can be calculated from

$$d_{j,t} = r_j \big(k_{j,t-1} + 0.5 * i_{j,t} \big). \tag{15}$$

Since the capital stock in national accounts is valued at replacement prices and not at historical acquisition costs as in firm's accounting, real depreciation must finally be inflated with the capital stock deflator $p_{j,t}^{K}$. Since this is only available until 2021, we update it with for 2022 and 2023 with the change of our own estimate of the investment deflator $p_{j,t}^{I}$.

$$d_{j,2022}^{nom} = d_{j,t} * \frac{p_{j,t}^{l}}{p_{j,2021}^{l}} * p_{j,2021}^{K}$$
(16)

Finally, we interpolate annual depreciation to the quarterly frequency using a temporal disaggregation method (Chow and Lin, 1976), with the quarterly depreciation for the total economy from sectoral accounts as regressor. This ensures that the aggregation of deprecation over industries fits with data from sectoral accounts for the total economy.

Net operating surplus

Finally, net operating surplus of firms is calculated as residual.

Adjusting compensation of employees and net operating surpluses for crisis subsidies

The subsidies granted during the COVID-19 crisis (mainly short-term work, sales compensation, loss bonus, fixed cost subsidies and loss compensation) are included in the income components compensation of employees and net operating surplus, distorting the inflation contributions of these components. Without further adjustments this leads to large positive contributions of net operating surplus and compensation of employees in

some quarters during the pandemic, as output fell and net operating surplus and compensation of employees remained high due to the subsidies, and a negative contribution of (negative) net taxes on production, which offset each other partly. It can be assumed that the COVID subsidies did not have a substantial impact on the pricesetting behaviour of firms since their goal was primarily to reduce losses and not to reduce prices. Therefore, net operating surpluses and compensation of employees (due to short-term work subsidies) were adjusted by the respective crisis subsidies.

Imports of goods and services

Data on imports of goods are available at the monthly frequency. We forecast missing observations (usually one month) with ARIMA time series models (usually one month is estimated to complete the current quarter). We use the same treatment for service imports, which are available at the quarterly frequency and where usually one quarter is missing when quarterly national accounts are released.

5.2 Calculation of consumption-implied cost components

The next step is to address the problem that industry-specific cost components in national accounts (value-added, imports, net taxes on intermediate goods) correspond to the sum of all final demand components and not consumption only.¹⁹ For example, a strong growth of equipment investment (which has a very high import content) causes imports to grow strongly. Without differentiating between various final demand components, this strong import growth results in an overestimation of the import contribution to consumer price inflation.

To avoid such distortions, we must calculate cost components that are specific to private consumption. We begin by calculating cost shares $CS_{k,f}^{IO}$ for all 11 demand components f in national accounts except changes in inventories.²⁰ We perform a multiplier analysis using the input-output table as described for private consumption in section 4 (see equations (4) and (5)). We then aggregate over the 74 CPA goods to obtain aggregated results for each final demand component.²¹ This gives us an 11x80 matrix of cost shares similar to figure 4. We then calculate *hypothetical cost components* $\tilde{CC}_{k,f,t}^{QNA}$ for each final demand component by multiplying the respective cost share ($CS_{k,f}^{IO}$) from the input-output table with the final demand component from QNA ($d_{f,t}^{QNA}$).

¹⁹ For taxes and subsidies on products, consumption-specific data can be obtained directly from input-output tables.

²⁰ Final consumption by households, final consumption by government, final consumption by NPISH, gross fixed capital formation - dwellings, gross fixed capital formation - other buildings and structures, gross fixed capital formation - machinery, gross fixed capital formation - transport equipment, gross fixed capital formation - cultivated assets, gross fixed capital formation - intangible fixed assets, exports of goods, exports of services.

²¹ Note that the results of the multiplier analysis for the different demand components are the same for each CPA-74 good. Differences in aggregated cost components between demand components are caused by differences in the goods structure.

$$\widetilde{c}c_{k,f,t}^{QNA} = cs_{k,f}^{IO} * d_{f,t}^{QNA}$$
(17)

Finally, we calculate *consumption-implied cost components* $(cc_{k,C,t}^{QNA})$ by multiplying the share of the hypothetical cost component of private consumption in the sum of the hypothetical cost components of all demand components with the cost component from QNA $(cc_{k,t}^{QNA})$.

$$cc_{k,C,t}^{QNA} = \frac{\widetilde{cc}_{k,C,t}^{QNA}}{\sum_{f=1}^{F} \widetilde{cc}_{k,f,t}^{QNA}} * cc_{k,t}^{QNA}$$
(18)

5.3 Update of cost shares from the input-output table with growth of consumption-implied cost components

As a next step, we update the cost shares from the input-output table with the growth of consumption-implied cost components $(cc_{k,i,t})$ in quarter *t*.

$$cs_{k,i,t} = cs_{k,i}^{IO45} * \frac{cc_{k,C,t}^{QNA}}{cc_{k,C,2019}^{QNA}}$$
(19)

These cost shares²² take structural shifts since the year of the input-output table such as the sharp increases of energy prices into account. The increasing importance of energy is reflected in a higher share of the energy cost component, resulting in a higher contribution than derived from the input-output table 2019.

5.4 Calculation of inflation contributions

In the final step, we use these cost components to calculate inflation contributions. We first have to calculate nominal private consumption of good i (which is not available in quarterly national accounts) as the sum of its cost shares.

$$c_{i,t} = \sum_{k=1}^{K} c s_{k,i,t} \tag{20}$$

Then we divide by the consumer price index to obtain real consumption of good *i*.

$$c_{i,t}^r = \frac{c_{i,t}}{p_{i,t}} \tag{21}$$

After this, we calculate unit cost shares (per real unit of consumption good *i*)

$$ucs_{k,i,t} = cs_{k,i,t} / c_{i,t}^r$$
(22)

Finally, we obtain contributions to year-on-year inflation using the same decomposition formulae as used for value-added deflator (equation (3)).

²² The results of (19) are no cost shares in a literal sense they do not add up to one. However, this does not matter for the decomposition since private consumption is calculated as the sum of all cost shares.

$$\frac{dp_{i,t}}{p_{i,t-4}} = \sum_{k=1}^{K} \frac{ducs_{k,i,t}}{ucs_{k,i,t-4}} \frac{ucs_{k,i,t-4}}{p_{i,t-4}}$$
(23)

6. Data and computational details

The decomposition requires various tables from the input-output statistics, aggregation matrices, HICP data and data from quarterly and annual national accounts. Table A-3 gives an overview of the data used for the decomposition. For Austria, we use the latest input-output table from published by the statistical office for year 2019 (Statistik Austria, 2023). The input-output table is available for 74 industries/goods. We use the symmetric input-output tables for domestic production and for imports at basic prices, the tables on wholesale, retail and transport margins at current prices, the tables on taxes and subsidies on production at current prices and the table on final use at purchasers' prices.

We conduct the decomposition of the HICP at the COICOP 3-digit level (45 goods). We also decompose headline inflation, inflation for special groupings (4) and divisions (12). Since the decomposition method is non-linear²³, we calculate the decomposition based on aggregated cost components instead of aggregating the results. We do the same for annual vs. quarterly results. Due to the publication lag of the first full release of quarterly national accounts of two months we can calculate our decomposition with a delay of one and a half months after the publication of HICP inflation of the last month of a quarter.

7. Interpretation of the results

Before presenting the results in the next section, we should look at the meaningfulness and the limitations of our approach. There are several points to consider.

7.1.1 The magnitude of inflation cannot be explained by the decomposition

By construction, the decomposition of the HICP – as well as the decomposition of the value-added deflator – cannot explain the magnitude of inflation, but just the distribution of incomes generated by the production of the consumption bundle underlying the HICP to the production factors.

7.1.2 Reasoning from an accounting identity

An important point to consider is that the decomposition is based on an accounting identity. It is well-known in economics that inferring a causal relationship from an identity can lead to misleading results (Albrecht, 2023).

²³ It is non-linear since we divide each cost component k for the production of consumption good i by private consumption good i, which is calculated as the sum of all cost components. Table A-4 shows the aggregation error when aggregating the results vs. the cost components for the quarterly and the annual decomposition.

Let us demonstrate this with the GDP demand side identity which states that GDP is the sum of private consumption, investment, government consumption and exports minus imports (Y=C+I+G+X-M). We can make two types of errors when drawing conclusions from this identity. The *first* is that causality can run in the opposite direction as the identity might suggest. Whilst the GDP demand side identity suggests that GDP is caused by the sum of the demand components less imports, the causality might run in the opposite direction in the case of a positive technology shock that increases the level of production, which in turn leads to higher demand. The second error is that we - often implicitly – make the ceteris-paribus assumption that the other variables on the righthand side of the identity do not change. This assumption might not hold in many cases. In our example drawing the conclusion that an increase of a demand side component (e.g. exports) by one euro leads to an increase of GDP by one euro is false for at least two reasons. Firstly, a demand shock causes GDP to increase only in the case of idle production capacities. Otherwise, export demand might crowd out other demand components. Secondly, exports have a direct import content which causes GDP to increase less than exports even in the case of idle production capacities.

Whilst the identity does not establish a causal relationship between the variables on the right-hand side and the variable on the left-hand side, it does also not rule out that such a relationship exists. Therefore, we need an economic theory in the background that establishes the causal relations between the variables to reason from an accounting identity.

7.1.3 Can the decomposition be informative about the nature of shocks that drive inflation?

Next, we want to look at the question whether the decomposition can tell us anything about the nature of shocks that drive inflation. In his survey on macroeconomic theories of inflation, Totonchi (2011) groups the drivers of inflation into five categories, namely monetary shocks, demand shocks, supply-side shocks, structural and institutional factors.²⁴ While it seems obvious that our decomposition cannot help identifying structural and institutional factors as the drivers of inflation, we want to look whether it can be informative about the remaining three inflation shocks. According to the *monetary theory of inflation*, money supply drives inflation. *Demand-pull inflation* occurs when the level of aggregate demand grows faster than the level of aggregate supply, which causes prices to increase. *Cost-push inflation* occurs when firms increase their prices when their costs rise. The typical method to disentangle the different types of shocks is to estimate an empirical model that includes theory-based restrictions such as a structural VAR model (Shapiro, 2022, Firat and Hao, 2023).

The mechanics of demand-pull and cost-push inflation are different. For *cost-push inflation*, the trigger is an initial increase of the costs of a production input, e.g. imported raw materials. As a response, firms try to increase their prices to maintain their profit

²⁴ The literature on macroeconomic theories of inflation is broad. Other contributions are Burton, 1972, Parkin, 1994, Schwarzer, 2018, Totonchi,2011, Fahlevi et al., 2020, just to name a few.

margins. In the next stage, employees demand higher wages to compensate for their loss of purchasing power. This results in further price increases.²⁵ Weber and Wasner (2023) have referred to this process as "sellers's inflation". If the initial trigger comes from abroad, then this should show up as a high contribution of imports²⁶. The domestic cost components should follow in the same order as outlined above, namely profits followed by wages. If we can observe such a pattern in our decomposition, then cost-push inflation is a likely explanation. For *demand-pull inflation*, we have to distinguish between a domestic and an international demand shock: For a domestic demand shock, imports should not have the most important contributions. With regard to domestic cost components, we would expect the same order as for a cost-push shock, namely that in the beginning firms' profits will benefit from inflation, followed by wages. For an international demand shock that drives up commodity prices, imports may also contribute significantly to domestic inflation. Through the lens of our decomposition, a *monetary shock* is indistinguishable from a *demand shock* without the use of additional indicators with regards to the monetary conditions.

7.1.4 The role of profits

Special care must be taken when analyzing the role of profits. Profits are not a costcomponent in a literal sense but could be understood as the residual between the revenues and costs of a firm.²⁷ With regards to the direction of causality, there are two possibilities. Firstly, firms set prices based on their costs plus a profit markup. Secondly, profits are caused by the difference between prices and the costs of the firm. The behavior of prices and profits depends on a variety of factors, such as the market structure, the degree of price rigidities, the price elasticity of demand and the type of shock. In addition to this more fundamental problem, the profit measure in National Accounts (net operating surplus) has to be treated with special care for a variety of reasons. Firstly, it is calculated as a residual by subtracting compensation of employees, taxes less subsidies and consumption of fixed capital from value added by industry. Hence, all measurement errors of the other variables impact on net operating surplus. This holds especially for consumption of fixed capital, which is a crude estimate calculated with the perpetual-inventory approach, utilizing assumptions on depreciation rates. Secondly, it contains not only firm's profits in a narrow sense, but also net mixed income (which incorporates the entrepreneurial income of unincorporated enterprises in the household sector as well as the labor compensation of self-employees). It also includes the corporate income tax and interest payments, which may have implications for economic analysis (ECB, 2004). Another important point to consider is that an increase of nominal (unit) profits does not necessarily imply that profitability (the mark-up or

²⁵ Ruch (2016) and Baba and Lee (2022) have estimated second-round effects of energy price shocks.

²⁶ Increasing import prices can be the consequence of a global demand shock, but we don't want to further explore this argument.

²⁷ Profits include interest on capital which are costs to the firm.

profit margin) has increased. Colonna et. al (2023) show that only an increase of the markup can signal price pressures arising from profits.²⁸

7.1.5 Further caveats

Besides the points discussed above, there are further caveats. Firstly, the analysis depends on the cost structure of the last year where an input-output table is available. If this is a year which is subject to a large shock, this could impede the results. Secondly, the analysis is static and assumes that all transactions within the production chain occur within one period. While this should be relatively innocuous at the annual frequency, quarterly results should be interpreted with caution. Thirdly, the inter-industry linkages reported in the input-output table do not have to apply in the short term, i.e., production can be maintained for some time without some of the intermediate inputs. Fourthly, the analysis hinges on the quality of quarterly national accounts data, which are often subject to revisions.

8. Results of the HICP decomposition for Austria

In this section, we present the results of the decomposition for Austria from the first quarter of 2019 until the first quarter of 2023 (figure 6). Inflation in Austria was very low until the end of 2020. From the last quarter of 2020 onwards, imports of non-energy goods began to push up consumer prices due to the supply-chain disruptions in the wake of the pandemic and the related price increases. The contribution of non-energy imports in the first quarter of 2021 of 4.1 percentage points was broad-based and mainly determined by imports of manufactured goods (1.6 pp), imports of machinery (0.5 pp), imports of crude materials (0.4 pp), imports of miscellaneous manufactured articles (0.4 pp), imports of crude materials (0.4 pp) and food (0.2 pp). Energy imports slightly dampened the inflation contribution of imports.

Prices for energy imports (mainly gas) started to rise even before the start of Russian war against the Ukraine in February 2022. The contributions of energy imports led to an increase in inflation from mid-2021 onwards. From the third quarter of 2021 until the fourth quarter of 2022, energy imports contributed 2.7 percentage points to headline inflation on average. For the year 2022, the contribution amounts to 2.5 percentage points or 29 % of inflation (+8.6 %). In the first quarter of 2023, energy imports slightly dampened inflation.

The role of profits for inflation is determined both by the pandemic and by increasing energy prices. During the pandemic, they fell stronger than real output and dampened prices. Profits of the energy sector began to contribute to inflation from the third quarter

²⁸ Other profit measures to consider are the profit share (profits over value-added), the mark-up (profits over unit labor costs), profit margins (profits over total output) and the rate of return on capital (profits over the capital stock) (ECB, 2004). In addition to these profit measures derived from National Accounts, profit measures based on firm's data can provide additional insights.

of 2022 onwards. This implies a lag of one year to energy imports. This is a very plausible result, since contracts for conducted energy (electricity, gas and district heating) are usually adjusted once a year in Austria. The increase in the contribution of profits in the non-energy sector is to a large extend driven by the recovery after the pandemic. The strong positive contribution of 3.6 percentage points in the first quarter of 2022 originates from accommodation and food service activities (5.8 pp). This was partly offset by lower profits in trade and repair of motor vehicles, manufacturing, and services. In the fourth quarter of 2022 and the first quarter of 2023, profits in the non-energy sector outpaced profits in the energy sector in their role for inflation. Accommodation and food service activities activities accounted for two thirds of the non-energy sectors contribution in both quarters. Additional contributions mainly came from finance and insurance activities and construction.



Figure 6: Inflation contributions to headline HICP inflation

Taxes less subsidies also contributed to inflation. The VAT reduction dampened inflation for four quarters starting in the third quarter of 2020. In 2022, the suspension of the reduction contributed positively to inflation. Note that wages and profits have been adjusted for crises-related subsidies, which also eliminates the contributions of these subsidies.

Due to the delayed wage adjustment process that is common in Austria²⁹, there has been no substantial price pressure from wages in 2022. Wages contributed 1.3 percentage points or 15 % to inflation, which is clearly below the average cost share of wages of 27% (2019). In the fourth quarter of 2022 and the first quarter of 2023, the collectively agreed wage increases began to accelerate and lead to slightly higher wage contributions.

²⁹ The Austrian wage bargaining is characterized by wage leadership reached by the metal workers, setting a precedent for other sectors to follow. The outcome are staggered wages, which usually remain valid for one year (Knell and Stiglbauer, 2009).

In addition to the decomposition of headline inflation, we present the decomposition of the four special groupings to demonstrate the granularity of the approach in figure 7. The profile and the structure of inflation differ considerably between the groups. *Food* inflation rose significantly during the course of 2022, reaching 13.7 % in the first quarter of 2023. This was to a large extend driven by food imports, which explain more than half of food inflation. Profits in trade and repair of motor vehicles, agriculture and the energy sector also contributed to food inflation. The role of wages was below their cost share in 2022.



Figure 7: Inflation contributions for special groupings

The inflation profile and the determinants for *non-energy industrial goods* are similar to food. Imports are also the most important component, with imports of manufactured goods contributing most to inflation. The domestic contribution (especially of wages) is slightly higher than for food inflation.

Energy inflation peaked at 48 % in the third quarter of 2022, with energy imports explaining the bulk of inflation. Profits in the energy sector explain 13 % of energy inflation in 2022. The role of the different energy imports differs over time. Gas imports contributed most to inflation the fourth quarter of 2021 and the first half of 2022. The contribution of oil imports was more or less stable at 8 percentage points in the course of

2022. In the first quarter of 2023, energy inflation fell substantially due to the declining role of energy imports.

Services inflation exhibits a completely different profile and structure. Compared to headline inflation, it is lower and somewhat delayed. Its composition was more affected by the pandemic than for the other groups. This shows up in the contributions of profits from the non-energy sector. In the course of 2022, the suspension of the VAT reduction fuelled inflation. In the first quarter of 2023, profits in accommodation and food service activities contributed most to inflation, explaining 4.8 percentage points of services inflation of 7.6 %. Profits in finance and insurance services contributed 0.8 percentage points. Interestingly, the contribution of wages to services inflation was negligible.

9. Explaining the differences of the decomposition results between value-added deflator and HICP

In this section we compare the decomposition of the HICP with the decomposition of the value-added deflator. The decompositions differ due to five factors (figure 8). First, imports play an important role for the HICP, while they are not included in the valueadded deflator. Second, taxes and subsidies on goods are not part of the value-added deflator.³⁰ The VAT reduction during the COVID-19 pandemic and its expiry as well as the reduction in energy taxes are therefore not included. Third, the sectoral compositions of the two price indices differ. The value-added deflator takes total value added of all sectors into account, while the HICP decomposition weights the sectors with their consumption shares from the IO analysis. As a result, public services, construction and manufacturing play a smaller role in the HICP. At the same time, the service sectors have a significantly higher weighting in the HICP. The differences in the cost shares of the HICP and the value-added deflator can be found in the right panel of figure 5. While compensation of employees and gross operating surplus (including consumption of fixed capital) have a similar share (26.9 % vs. 27.7 %) for the HIPC, the share of compensation of employees (54.3 %) is much higher than the share of gross operating surplus (43.8 %) for the value-added deflator. This results from the differences in the sectoral composition of the two price indices.

 $^{^{30}\,}$ Taxes less subsidies on production are part of the value-added deflator.



Figure 8: Comparison of the decomposition of the value-added deflator and the HICP

Fourth, the series ultimately used in the decompositions differ. Whilst we use exactly the same time series as input in the calculations, we proceed one additional step for the HICP-decomposition. This is the calculation of the consumption-implied cost components (section 5.2). The consumption-implied cost components show a different course than the original national accounts series, which are used for the deflator decomposition. Specifically, private consumption fell more sharply during the pandemic than the other demand components and grew more strongly during the upswing. The fifth factor is given by the fact that we use a different quantity to calculate the unit cost components. In the case of the value-added deflator, we divide by real value added. For the HICP, we divide by the real consumption good. Consequently, the profile of a unit cost component can differ between the two decompositions, even if the nominal cost component is identical.

Besides imports and taxes less subsidies on products, these factors result in huge differences regarding the relative role of wages and profits. Wages play a much stronger role in the decomposition of the value-added deflator. The main reason is the sectoral composition (see above). The weight of public services (which have a very high wage share) in the HICP is considerably lower than their share in total value added.

Another striking difference between the two decompositions is the opposite sign of the contribution of profits in 2021. For the value-added deflator, the contribution is positive, while it has a negative sign for the HICP. The reason is that real consumption grew stronger than real value added in that year, leading to the opposite sign of changes (and hence contributions) in unit profits.

10. Summary and conclusions

In this paper we have proposed a novel methodology for a production-side decomposition of HICP inflation based on national accounts data. By combining input-output data with quarterly national accounts data, HICP inflation can be decomposed at a granular level. We used the approach to analyze the inflation developments in Austria for the period from the first quarter of 2019 to the first quarter of 2023. In 2022, energy as well as nonenergy imports were the most important contributors to inflation. Profits contributed to inflation from the second half of 2022 onwards, whilst there have been no substantial contributions from wages. We find that there are considerable differences in the inflation determinants between subindices of the HICP. Whilst imports played the main role for inflation of food, non-energy industrial goods and energy, they hardly contributed to services inflation. The results also clearly demonstrate how particularities like the lagged wage-bargaining process in Austria impact on inflation.

Compared to the results of the simple decomposition of the value-added deflator (which is often used as a proxy for the decomposition of consumer prices due to its simplicity), our approach has advantage that it provides an accurate picture of consumer price inflation. This includes the contributions of imports, taxes less subsidies on products (including the value-added tax) and detailed contributions of cost components by industries.

The results have a high policy relevance. For central banks, they allow to disentangle domestic and foreign determinants of inflation, which is of high relevance for monetary policy. They also can shed light on the role of distributional aspects between wages and profits and can provide valuable input into the annual wage bargaining rounds. Due to the breakdown of cost components by industry, they allow to discuss policy questions such as the role of sectoral profits for inflation in detail.

There are several avenues for *future research* in this area. Given the crucial importance of inflation for central banks, other central banks should aim to implement such a decomposition for their countries. If one wants to dig deeper and split up the contributions of unit cost components (e.g. unit labor costs) into a price and a productivity component, this can be done easily for cost components where price data are available (e.g. wages). Another way forward would be to decompose the contribution of a cost component into a "normal" component and an "excessive" component, which results from above-average growth of a cost component. This allows to better assess distributional aspects. Summing up, our decomposition approach provides new valuable insights and should not be missing in the toolboxes of central banks and other policy institutions.

11. References

- Abberger, K. and W. Nierhaus (2023): Die Preisentwicklung der inländischen Wertschöpfung. Zum Anstieg des BWS-Deflators im Jahr 2022. ifo Schnelldienst 5/2023, 47-53.
- Arce, O. E. Hahn and G. Koester (2023): How tit-for-tat inflation can make everyone poorer. The ECB blog, 30 March 2023.
- Aucremanne, L. and R. Wouters (1999): A Structural VAR Approach to Core Inflation and its Relevance for Monetary Policy. In Measures of Underlying Inflation and Their Role in the Conduct of Monetary Policy: Proceedings of the Workshop of Central Bank Model Builders, 1-44. Basel: Bank for International Settlements.
- Baba, C. and J. Lee (2022): Second-round effects of oil price shocks Implications for Europe's inflation outlook. IMF Working Paper WP/22/173.
- Burton, J. (1972): Theories of Inflation. In: Wage Inflation. Macmillan Studies in Economics. London: Palgrave.
- Byrne, S., D. McLaughlin and M. O'Brien (2022): Business costs and consumer price inflation in Ireland. Central Bank of Ireland Economic Letter, Vol. 20222, No. 3.
- Cai, M. and T. Vandyck (2020): Bridging between economy-wide activity and household level consumption data. Matrices for European countries. Elsevier: Data in Brief 30(2020), 1-4.
- Chow, G. and A. L. Lin (1976): Best linear unbiased estimation of missing observations in economic time series. Journal of the American Statistical Association, 71, 719-21.
- Colonna, F., R. Torrini and E. Viviano (2023): The profit share and firm mark-up: how to interpret them? Banca d'Italia Occasional Papers 770.
- Dhingra, A. (2023): A cost-of-living crisis: Inflation during an unprecedented terms of trade shock. Speech, given at the Resolution Foundation. Downloaded from <u>https://www.bankofengland.co.uk/-/media/boe/files/speech/2023/march/a-cost-of-living-crisis-speech-by-swati-dhingra.pdf</u>, January 11th, 2024.
- ECB (2003): Inflation differentials in the euro area: potential causes and policy implications. ISBN 92-9181-411-3
- ECB (2004): Measuring and analysing profit developments in the euro area. ECB Monthly Bulletin January 2004, 63-73.
- ECB (2006): Using national accounts data to gauge price pressure in the euro area. Box 6 in ECB Monthly Bulletin December 2006, 52-55.
- Eder, A., W. Koller and B. Mahlberg (2019): Price competitiveness in the European Monetary Union: A decomposition of inflation differentials based on the Leontief Input-Output price model for the period 2000 to 2014. Mimeo.
- European Commission (2023): Inflation differentials in Europe and implications for competitiveness. Thematic note to supper in-depth reviews. Institutional paper 198, April 2023.
- Fahlevi R., R. Ernayani, W. Lestari, A. Hubur, and A. Whyudi (2020): A brief review on the theory of inflation. Journal of Critical Reviews, Vol 7(8), 2069-2076.
- Firat, M. and O. Hao (2023): Demand vs. supply decomposition of inflation: Cross-country evidence with applications. International Monetary Fund Working Paper WP 23/205.

- Fritzer, F., L. Reiss and M. Schneider (2023): The role of profits in the development of domestic price pressure in Austria- forthcoming in OeNB: Monetary Policy & the Economy.
- Guo S., P. Karam and J. Vlcek (2019): Decomposing inflation dynamics in the Philippines. IMF Working Paper 19/153.
- Hahn, E. (2019): How do profits shape domestic price pressures in the euro are. ECB Economic Monthly Bulletin, Issue 6/2019, 65-68.
- Hansen, N.-J., F. Toscani and J. Zhou (2023): Euro area inflation after the pandemic and energy shock: import prices, profits and wages. IMF Working Paper 23/131.
- Haskel, J. (2023): What's driving inflation: wages, profits, or energy prices? speech by Jonathan Haskel given at the Peterson Institute for International Economics, Washington DC, 25 May 2023. Downloaded from <u>https://www.bankofengland.co.uk/-/media/boe/files/speech/2023/may/whats-driving-inflation-wages-profits-or-energyprices-speech-by-jonathan-haskel.pdf</u>, June 24th, 2023.
- Hebbing, G. and B. Öztürk (2023): The contribution of profits and wages to Dutch inflation. De Nederlandsche Bank Analysis.
- Huber, E. (2015): Schätzung des Kapitalstocks in der österreichischen VGR. Konzepte, Methoden und Ergebnisse. Statistik Austria: Statistische Nachrichten 6/2015, 476-81.
- Jaumotte, F. and H. Morsy (2012): Determinants of inflation in the euro area: the role of labor and product market institutions. IMF Working Paper No. 12/37.
- Kambar G. and B. Wong (2018): Global factors and trend inflation. BIS Working Paper 688.
- Knell, M. and A. Stiglbauer (2009): Wage staggering and wage leadership in Austria review and implications. Monetary Policy & The Economy Q4/09, 79-97.
- Martel, S. (2008): A structural VAR approach to core inflation in Canada. Bank of Canada Discussion Paper 2008-10.
- Moretti, L., L. Onorante and S.Z. Saber (2019): Phillips Curves in the Euro Area. ECB Working Paper No. 2295.
- Parkin, M. (ed.) (1994): The theory of inflation. Edward Elgar Publishing.
- Richardson, D., M. Saunders and R. Denniss (2022): Are wages or profits driving Australia's inflation. An analysis of National Accounts. The Australian Institute Discussion Paper, July 2022.
- Ruch, F.-U. (2016): Second-round effects on inflation and underlying inflation. Dissertation at the University of Stellenbosch, South Africa.
- Rumler, F. (2007): Estimates of the Open Economy New Keynesian Phillips Curve for Euro Area Countries. Open Economies Review 18 (4). 427–451.
- Schwarzer, J. A. (2018): Cost-Push and Demand-Pull Inflation: Milton Friedman and the "Cruel Dilemma". Journal of Economic Perspectives Vol 32(1), 195-210.
- Shapiro, A. H. (2022): Decomposing supply and demand driven inflation. Federal Reserve Bank of San Francisco Working Paper 2022-18.
- Smets F, and R. Wouters (2003): An Estimated Stochastic General Equilibrium Model of the Euro Area. Journal of the European Economic Association 1. 1123–1175.

- Statistik Austria (2023): Input-Output-Tabelle 2019. Inklusive Aufkommens- und Verwendungstabelle. Wien: Statistik Austria. Available from https://www.statistik.at/fileadmin/user_upload/Input-Output-Tabelle-2019.pdf.
- Stock, J.H. and M.W. Watson (2019): Slack and Cyclically Sensitive Inflation. National Bureau of Economic Research Working Paper 27487.
- Totonchi, J. (2011): Macroeconomic Theories of Inflation. 2011 International Conference on Economics and Finance Research IPEDR vol. 4, 459-462.
- Weber, I. and E. Wasner (2023): Sellers' Inflation, Profits and Conflict. Why can Large Firms Hike Prices in an Emergency? University of Massachusetts Amherst: Economic Department Working Paper Series 2023-3.
- Wehinger, G. D. (2000): Causes of Inflation in Europe, the United States and Japan: Some Lessons for Maintaining Price Stability in the EMU from a Structural VAR Approach. Empirica Kluwer Academic Publishers, 2000: 83-107.

12. Appendix

A-1 Aggregation

Aggregation between different statistical classifications is a crucial part of this exercise. The best way to aggregate matrices or vectors is to set up an appropriate aggregation matrix for each aggregation. Therefore, it is important to understand the difference between two different aggregation matrices, which we name *assignment matrices* and *weight matrices*.

Assignment matrices are used for aggregations where variables in levels (e.g. imports in MEUR) are aggregated from a more granular to a less granular level (e.g. CPA-74 to SITC/BPM6-25). In this case, a CPA-74-import good is usually assigned to one SITC/BPM6-25 category by setting the value of the corresponding element of the assignment matrix to one. However, there can be occasions where one CPA-74 good is assigned to two or more SITC/BPM6-25 goods. In this case, the respective entries in the aggregation matrix must sum up to one. In any case, the row sum of an assignment matrix. Note that the sum of the aggregated vector equals the sum of the original vector.

Figure A-1: Types of aggregation matrices



Weight matrices are a generalisation of weight vectors (e.g. HICP weights). The multiplication of a vector with a weight delivers a scalar number. *Their column sum of a weight matrix must be equal to one*, implying that the aggregate is a weighted sum of its components. We use these weight matrices to aggregate between different levels of the COICOP-classification. The right part of figure A-1 shows an aggregation with a weight matrix. Note that the mean sum of the aggregated vector does not have to equal the mean of the original vector.

A-2 Tables

Table A-1: Correspondence table from CPA-74 to COICOP-45 at purchasing prices

Table 25

Final consumption expenditure by households ÖCPA x COICOP

current prices, in 1000 € (at purchasing prices)

		01.1	01.2	02.1	02.2	03.1	03.2	 12.5	12.6	12.7
	ÖCPA x COICOP	Food	Non- alcoholic beverages	Alcoholic beverages	Tobacco	Clothing	Footwear	 Insurance	Financial services n.e.c.	Other services n.e.c.
01	Products of agriculture, hunting and related services	2772043	-	-	70946	-	-	 -	-	-
02	Products of forestry, logging and related services	-	-	-	-	-	-	 -		-
03	Fish and fishing products	166926	-	-	-				-	-
05-07	Coal a.lignite; crude petroleum a.natural gas; metal ores	-	-	-	-	-	-	 -	-	-
08-09	Other mining a. quarrying prod.; mining support services	-	-	-	-	-			-	-
45	Wholesale- a. retail trade, repair of motor vehicles	-	-	-	-	-	-	 -	-	-
46	Wholesale trade, exc. o.motor vehicles acycles	-	-	-	-	-	-	 -	-	-
47	Retail trade, exc. o.motor vehicles acycles	-	-	-	-	-	-	 -	-	-
49	Land transport services a. transport services via pipelines	-	-	-	-	-	-	 -	-	-
50	Water transport services	-	-	-	-	-	-	 -	-	-
51	Air transport services	-	-	-	-	-	-	-	-	-
52	Warehousing and support services for transportation	-	-	-	-	-	-	-	-	-
53	Postal and courier services	-	-	-	-	-	-	 -	-	-
65	Insurance, reinsurance and pension funding services	-	-	-	-	-	-	 5756736		-
		-	-	-	-	-		 -	333177	-
93	Sporting services, amusement and recreation services	-	-	-	-	-	-	 -	-	-
94	Services furnished by membership organisations	-	-	-	-	-			-	113487
95	Repair services of computers, pers. a. household goods	-	-	-	-	88029	62459	-		-
96	Other personal services	-	-	-	-	94614	-	-	-	1100299
97	Services of households as employers of dom. personnel	-	-	-	-	-	-	-	-	-
	Total	17680532	2150774	2722891	3789421	9181600	2218204	6073872	2500793	2108461

S: STATISTICS AUSTRIA

Table A-2: Correspondence table from CPA-74 to COICOP-45 at basic prices

Table 25

Final consumption expenditure by households ÖCPA x COICOP

current prices, in 1000 € (at basic prices)

		01.1	01.2	02.1	02.2	03.1	03.2	 12.5	12.6	12.7
	ÖCPA x COICOP	Food	Non- alcoholic beverages	Alcoholic beverages	Tobacco	Clothing	Footwear	 Insurance	Financial services n.e.c.	Other services n.e.c.
01	Products of agriculture, hunting and related services	1766450	-	-	45209	-	-	 -	-	-
02	Products of forestry, logging and related services	-	-	-	-	-		 -	-	-
03	Fish and fishing products	104431	-	-	-	-		 -	-	-
05-07	Coal a.lignite; crude petroleum a.natural gas; metal ores	-	-	-	-	-		 -	-	-
08-09	Other mining a. quarrying prod.; mining support services	-	-	-	-	-	-	 -	-	-
45	Wholesale- a. retail trade, repair of motor vehicles	170839	20199	24781	35637	93800	22449	-	-	-
46	Wholesale trade, exc. o.motor vehicles acycles	1302219	185791	277745	392277	261937	49991	-	-	-
47	Retail trade, exc. o.motor vehicles acycles	4329941	458250	478157	699666	3141853	773362	-	-	-
49	Land transport services a. transport services via pipelines	79624	6142	3079	5240	13302	2551	-	-	-
50	Water transport services	2748	212	106	181	459	88	-	-	-
51	Air transport services	1412	109	55	93	236	45	-	-	-
52	Warehousing and support services for transportation	34727	2679	1343	2285	5802	1113	-	-	-
53	Postal and courier services	-	-	-	-	-	-	-	-	-
65	Insurance, reinsurance and pension funding services	2200	170	85	145	368	70	5756736	-	-
93	Sporting services, amusement and recreation services	-	-	-	-	-	-	-	-	-
94	Services furnished by membership organisations	-	-	-	-	-	-	-	-	113487
95	Repair services of computers, pers. a. household goods	-	-	-	-	88029	62459	-	-	-
96	Other personal services	-	-	-	-	94614	-	-	-	1100299
97	Services of households as employers of dom. personnel	-	-	-	-	-		-		-
	Total	2150774	2150774	2722891	3789421	9181600	2218204	6073872	2500793	2108461

ů	1001	Data	Init of	Erocitonoli	Dimoneton	Commont
õ	Inel	Data	measure	Leduency		
1	hut output	lata (data_lO.xlsx)				
č	onsumption data	Private constitution incl. Net taxes and marchins	Nominal	٩	74x8	
4	b05	Final demand component at nurchasing prices	Nominal	4	74x10	
π	h16	Wholesale trade marrins final demand	Nominal	4	74×10	
s a	514 714	Retail trade marnins final demand	Nominal	< ⊲	74×10	
5 0	818	Trapenort margine final demand	Nominal	(<	74/10	
ה מ	010	Product faves final demand	Nominal	(4	74×10	
π	h20	Subsidies of numbers final demand	Nominal	4	74×10	
ນ ແ	b25	Mapping matrix ÖCPA 74 x COICOP 45	Nominal	< ◄	74x45	
e a	b29	Input-output table, domestic production	Nominal	٩	74x74	
a)	p30	Input-output table, imports	Nominal	4	74x74	
◄	ggregtion m	atrices (data_aggmat.xlsx)				
			ç			
さん	4X13	Aggregation matrix from 74 NACE industries of input-output table to 13 industries of qu	Larte IU cost comp	onents (levels)	74×13	Assignment matrix
: # ;	5x12	Aggregation matrix from NACE /4 goods to 25 SH C/BP/WG goods (for impons or inter Aggregation matrix from 45 COICOP 3-digit goods to 12 COICOP divisions		ermediate good	45x12	Assignment naunx Weight matrix
4	5x4	Aggregation matrix from 45 COICOP 3-digit goods to 4 COICOP special aggregates			45x4	Weight matrix
ð	luarterly data	i data_QNA.xlsx)				
Ŧ	ICP data					
	ICP-M	HICP per COICOP good	Index	ΣZ	Tx46 Tv46	Headline and 45 COICOP 3-digit groups
r =	ICP.W-A	HICP at constant tax hates per COLOCP good HICP weights	nuex 1	N A	1 X40 T X45	Needed to calculate the imitation continuous on taxes on products
ā	uarterly cost co	n ponents				
· ~	1G.R-Q	Gross value added per industry	Real	ø	Tx13	For decomposition of the value-added deflators; not needed for decomposition of HICP
à	1G-Q	Gross value added per industry	Nominal	ø	Tx13	
ò	å	Compensation of employees	Nominal	Ø	Tx13	
n i	x9X39.T-Q	Taxes less subsidies on production per industry - total	Nominal	Ø	Tx13	Author's own compilation based on auxiliary data (tax data, data on subsidies)
-i é	N-N-28-28-20-20-20-20-20-20-20-20-20-20-20-20-20-	Taxes less subsidies on production per industry - crisis-related wage subsidies (snort- Taxes less subsidies en andustion par industry andustry andustry andustry	term Nominal	3 0	TX13	
	K9X39 R-O	r axes ress subsidies on production per industry - crisis-related prom subsidies Taxes less subsidies on production per industry - other taxes less subsidies	Nominal	90	Tx13	
	31-Q	Subsidies on products per COICOP per good	Nominal	a	Tx45	Author's own compilation based on auxiliary data
~	21x31.ID-Q	Taxes less subsidies on intermediate goods	Nominal	ø	Tx1	Proxied by overall taxes less subsidies
1 C C	71-M	Imports of commodieties	Nominal	Σ	Tx13	Level of disaggregation can be choosen (1)
6	72-Q	Imports of services	Nominal	ø	Tx12	Level of disaggregation can be choosen (1)
័	uarterly data us	ed to calculate consumption-caused cost components				
õ	emand-Q	Final demand components	Real	σ	Tx11	
	nnual cata used	for cross-checking				
	1G-A	Gross value added per industry	Nominal	٩	Tx13	Not necessary, but we have used different sources for quarterly (Statistics Austria) and annual (Eurostat) data
- m	1G.R-A	Gross value added per industry	Nominal	۷	Tx13	
à	1G-A	Gross value added per industry	Nominal	A	Tx13	
ò	1-A	Compensation of employees	Nominal	۷	Tx13	and March 2010
ñ	x9X39-A	Taxes less subsidies on production per industry - total	Nominal	A	Tx13	
ຈັ ໄ	51C-A	Consumption of fixed capital	Nominal	∢ <	1×13	Important for cross-check, since quarterly data has been estimated
đ	NCHZ		NOTIFIAL	¢	CIXI	iriportarit tot cross-crieck, since quarterly data rias peeri carculated as residual
ő	ata used for cal	ulation of consumption of fixed capital (perpetual inventory)				
1	EP-A	Consumption of fixed capital per industriy	Nominal	۷	Tx13	
3	SR-A	Capital stock per industry	Nominal	A	Tx13	
ごi	SN-A	Capital stock per industry	Real	۷.	Tx13	
— i	R-A	Gross fixed capital formation per industry	Real	4 •	Tx13	Historical data until 2021 has been projected until 2024
⊢ 7	D-A	Gross fixed capital formation per industry	Deflator	< ۷	Tx13	Historical data until 2021 has been projected until 2024
ה ב	EP.101-0	Quarterly consumption of fixed capital for total economy		י כ		Quarterity profile used to interpolate annual data to quarters with Chow-Lin method
ĭ	Drec	Forecast of GFC and GFC deflator per industry	Real/Deliator	4	EIX13	Author's own torecast based on aggregate binPE ligures & judgement on sectoral distribution

Table A-3: Data needed for the decomposition

Table A-4: Difference of the aggregation of quarterly contributions vs. aggregation of the cost components to the annual frequency (aggregation error)

	HICP	Compensation	Taxes less	Consumptio	Net operating	Net operating	Energy	Other
		of employees	subsidies	n of fixed	surplus	surplus other	imports	imports
				capital	energy sector	sectors		
2022	0.0	-0.2	-0.1	-0.3	0.2	0.3	0.3	-0.3
2021	0.0	0.0	0.1	0.2	0.0	-0.9	0.2	0.3
2020	0.0	0.0	0.0	-0.1	0.0	0.4	0.0	-0.2

Source: Author's own calculations.

Index of Working Papers:

January 13, 2021	Maximilian Böck, Martin Feldkircher, Burkhard Raunig	233	A View from Outside: Sovereign CDS Volatility as an Indicator of Economic Uncertainty
May 20, 2021	Burkhard Raunig	234	Economic Policy Uncertainty and Stock Market Volatility: A Causality Check
July 8, 2021	Thomas Breuer, Martin Summer, Branko Uroševic	235	Bank Solvency Stress Tests with Fire Sales
December 14, 2021	Michael Sigmund, Kevin Zimmermann	236	Determinants of Contingent Convertible Bond Coupon Rates of Banks: An Empirical Analysis
February 14, 2022	Elisabeth Beckmann, Christa Hainz, Sarah Reiter	237	Third-Party Loan Guarantees: Measuring Literacy and its Effect on Financial Decisions
February 16, 2022	Markus Knell, Reinhard Koman	238	Pension Entitlements and Net Wealth in Austria
May 9, 2022	Nicolás Albacete, Pirmin Fessler, Peter Lindner	239	The Wealth Distribution and Redistributive Preferences: Evidence from a Randomized Survey Experiment
June 20, 2022	Erwan Gautier, Cristina Conflitti, Riemer P. Faber, Brian Fabo, Ludmila Fadejeva, Valentin Jouvanceau, Jan-Oliver Menz, Teresa Messner, Pavlos Petroulas, Pau Roldan-Blanco, Fabio Rumler, Sergio Santoro, Elisabeth Wieland, Hélène Zimmer	240	New Facts on Consumer Price Rigidity in the Euro Area
June 29, 2022	Svetlana Abramova, Rainer Böhme, Helmut Elsinger, Helmut Stix	241	What can CBDC designers learn from asking potential users? Results from a survey of Austrian residents

July 1, 2022	Marcel Barmeier	242	The new normal: bank lending and negative interest rates in Austria
July 14, 2022	Pavel Ciaian, Andrej Cupak, Pirmin Fessler, d'Artis Kancs	243	Environmental-Social-Governance Preferences and Investments in Crypto- Assets
October 18, 2022	Burkhard Raunig, Michael Sigmund	244	The ECB Single Supervisory Mechanism: Effects on Bank Performance and Capital Requirements
April 5, 2023	Norbert Ernst, Michael Sigmund	245	Are zombie firms really contagious?
May 8, 2023	Richard Sellner, Nico Pintar, Norbert Ernst	246	Resource Misallocation and TFP Gap Development in Austria
September 5, 2023	Katharina Allinger, Fabio Rumler	247	Inflation Expectations in CESEE: The Role of Sentiment and Experiences
October 16, 2023	Pietro Saggese, Esther Segalla, Michael Sigmund, Burkhard Raunig, Felix Zangerl, Bernhard Haslhofer	248	Assessing the Solvency of Virtual Asset Service Providers: Are Current Standards Sufficient?
October 20, 2023	Pirmin Fessler, Severin Rapp	249	The subjective wealth distribution: How it arises and why it matters to inform policy?
October 27, 2023	Burkhard Raunig, Michael Sigmund	250	Watching over 21,000 Billion Euros: Does the ECB Single Supervisory Mechanism Affect Bank Competition in the Euro Area?
December 5, 2023	Markus Knell	251	Housing and the secular decline in real interest rates
December 14, 2023	Niko Hauzenberger, Florian Huber, Thomas O. Zörner	252	Hawks vs. Doves: ECB's Monetary Policy in Light of the Fed's Policy Stance
February 28, 2024	Lukas Olbrich, Elisabeth Beckmann, Joseph W. Sakshaug	253	Multivariate assessment of interviewer- related errors in a cross-national economic survey

March 5, 2024	Nicolas Albacete, Pirmin Fessler, Atanas Pekanov	254	The Role of MPC Heterogeneity for Fiscal and Monetary Policy in the Euro Area
March 11, 2024	Martin Schneider	255	What contributes to consumer price inflation? A novel decomposition framework with an application to Austria